Author's response to reviews

Title: Laparoscopic sacrohysteropexy and myomectomy for uterine prolapse: a case report and review of the literature

Authors:

Radwan Faraj (faraj68@hotmail.com)
Jon Broome (jon.broome@rbh.nhs.uk)

Version: 7 Date: 2 November 2009

Author's response to reviews:

Dear Sir

As requested, revised manuscript has been submitted. Authors names been amended as first name, surname (no initials) Nationality of the patient been added Picture formatting and labelling also been amended as (1A,1B,1C,1D)

Many thanks
Kind REGards

Dr Radwan Faraj